Author’s response to reviews

Title: Syphilis-attributable adverse pregnancy outcomes in China: a retrospective cohort analysis of 1187 pregnant women with different syphilis treatment

Authors:

Huihui Liu (liuhh099@163.com)
Niannian Chen (cnnfezx@163.com)
Jia Yu (yjbigcs@163.com)
Weiming Tang (weimingtangscience@gmail.com)
Jianrong He (hjr0703@163.com)
Huiyun Xiao (xiaohuiyun1982@126.com)
Suifang Lin (lsfbigcs@163.com)
Fang Hu (hufang0351083@126.com)
Qiong Feng (fqfezx@163.com)
Joseph Tucker (jdtucker@med.unc.edu)
Huimin Xia (huimin.xia876001@gmail.com)
Xiu Qiu (qxiu0161@163.com)

Version: 5 Date: 08 Feb 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your meticulousness.

We have now clarified email address of Dr. Huimin Xia and provided a header for "Declaration" section in this manuscript.
We hope that the revised manuscript is now suitable for publication.

Yours sincerely,

Xiu Qiu